Author’s response to reviews

Title: Nuclear expression of p65 (RelA) in patients receiving post-operative radiotherapy for locally advanced squamous cell carcinoma of the head and neck

Authors:

Dirk Rades (rades.dirk@gmx.net)
Stefan Huttenlocher (stefan.huttenlocher@uksh.de)
Nina D Seibold (nina.seibold@uksh.de)
Maximilian P Gebhard (maximilian.gebhard@uksh.de)
Christoph Thorns (christoph.thorns@uksh.de)
Katrin Hasselbacher (katrin.hasselbacher@uksh.de)
Barbara Wollenberg (barbara.wollenberg@uksh.de)
Steven E Schild (sschild@mayo.edu)

Version: 4 Date: 16 December 2014

Author’s response to reviews:

Dear Editor,

Attached, please find our re-revised manuscript. The following revisions were made according to your suggestions.

1. T4 has been corrected in all 4 tables.
2. A Figure (new Figure 1) has been added showing an example of p65 staining (Legend to Figure 1, lines 338-339 + page 4, lines 113-114).
3. A paragraph regarding the HPV typing has been added (page 6, lines 153-158).
4. The sites of the HPV-positive tumors are now stated (page 6, lines 171-172).

We hope that this re-revised manuscript meets with your approval for publication.

Sincerely,

Dirk Rades (on behalf of the authors)